Author's response to reviews

Title: Serological evidence for human cystic echinococcosis in Slovenia

Authors:

Jernej Logar (jernej.logar@mf.uni-lj.si)
Barbara Soba (barbara.soba@mf.uni-lj.si)
Tadeja Kotar (tadejakotar@hotmail.com)

Version: 5 Date: 5 May 2008

Author's response to reviews:

Dear BMC Editorial Production Team

We are glad that manuscript will be published in BMC Infectious Diseases journal. Please find our formatted changes as follows:

Authors list; the authors do not have initials

Author's contributions was corrected to Authors' contributions

In the Acknowledgements, the sentence was added; The funding body had no role in the study design, collection, analysis or interpretation of data, or in the decision to submit the manuscript for publication.

With best wishes,

Jernej Logar